Author's response to reviews

Title: Care-seeking behavior and out-of-pocket expenditure for sick newborns among urban poor in Lucknow, northern India: a prospective follow-up study

Authors:

Neeraj M Srivastava (neersri@rediffmail.com)
Shally Awasthi (sawasthi@sancharnet.in)
Girdhar G Agarwal (girdhar1751@yahoo.co.in)

Version: 3 Date: 26 March 2009

Author's response to reviews: see over
Response to comments on revisions:

Major revisions (We require the author to make these changes)

(1) Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Response: We have now written the names of the authors as submitted in the system. In previous version it was “Neeraj Mohan Srivastava, Shally Awasthi, Girdhar Gopal Agarwal”. Now it is “Neeraj M Srivastava, Shally Awasthi, Girdhar G Agarwal”

(2) Conclusions - Please insert a Conclusions section after the Discussion. We recommend including a Conclusions heading above the paragraph which begins ‘In conclusion’.

Response: As recommended, we have inserted a Conclusion heading at the start of the last paragraph of the manuscript, as indicated.

(3) Authors’ contributions - Please indicate that all authors read and approved the final manuscript.

Response: We have now indicated this in the section of author’s contributions, and it reads: “Authors’ contributions: SA conceptualized the study and supervised the data collection, analysis and manuscript writing process. NMS collected the data. GA and NMS analyzed the data. All authors helped in the preparation of manuscript. All authors read and approved the final manuscript.”

(4) Reference 16 - please provide full details for this reference. This means for an Article providing volume/issue/page numbers and for a book the publisher/city.

Response: We have now also included the URL from where this reference was retrieved giving the full details for this reference:

It now reads:

Response: We have now given full details (included page numbers) for the above reference and it now reads:

Response: We are now giving Table 1 as additional file 1.

Response: As suggested, we have merged the tables into one larger complete table.

Response: In the authors’ affiliation we have now used sentence case i.e. India instead of INDIA (previous version).